Author's response to reviews

Title: A bibliometric search of citation classics in anesthesiology

Authors:

Ravi S Tripathi (ravi.tripathi@osumc.edu)
James M Blum (jmblum@med.umich.edu)
Thomas J Papadimos (thomas.papadimos@osumc.edu)
Andrew L Rosenberg (arosen@med.umich.edu)

Version: 4 Date: 30 October 2011

Author's response to reviews:

The authors names were corrected in the manuscript and listed in the format: First name Initial Last name. They author details were are also verified as correct.